Author's response to reviews

Title: The microbiome of chronic rhinosinusitis: culture, molecular diagnostics and biofilm detection

Authors:

Sam Boase (samboase@gmail.com)
Andrew Foreman (aforeman@adam.com.au)
Lorwai Tan (lorwai.tan@adelaide.edu.au)
Rachael Melton-Kreft (RKREFT1@wpahs.org)
Harshita Pant (mail@hpant.com)
Fen Z Hu (fhu@wpahs.org)
Garth D Ehrlich (GEHRLICH@wpahs.org)
Peter-John Wormald (peterj.wormald@adelaide.edu.au)

Version: 3 Date: 28 April 2013

Author's response to reviews: see over
To the Editor,

Thank you for considering our revised manuscript entitled “The microbiome of chronic rhinosinusitis: culture, molecular diagnostics and biofilm detection” We sincerely appreciate the constructive criticism from the reviewers and have addressed all recommendations below.

Reviewer 1 / Editor: We would like to request that you address the final point from Reviewer 1 regarding supplementary data.

We have included this table (Supplementary Table 1) which is probably suitable for online publication.

Editor: Include a paragraph on the limitations of your study.

We have included the study limitations in the final paragraph of the manuscript.

Thank you for your consideration,

Sincerely,

Prof PJ Wormald
Dr SJ Boase
Dr A Foreman
Dr E Cleland
Dr LW Tan
Ms R Melton-Kreft
Dr H Pant
Dr FZ Hu
Prof GD Ehrlich

Department of Otolaryngology, Head and Neck Surgery, University of Adelaide
Adelaide, South Australia
AUSTRALIA

Center for Genomic Sciences
Allegheny-Singer Research Institute
Pittsburgh, Pennsylvania, USA